DASSAULT SYSTEMES CUSTOMER LICENSE AND ONLINE SERVICES AGREEMENT

This Customer License and Online Services Agreement is made by and between the customer ("Customer") and the Dassault Systèmes legal entity ("3DS" or "DS"), as both are identified in the applicable Transaction Document. This Agreement is accepted by Customer by signing or otherwise accepting a Transaction Document referencing this Agreement or by other electronic means permitted by 3DS, such as click-to-accept and electronic signature. The parties agree as follows:

GENERAL TERMS

1. Definitions

3DS Group Company (or DS Group Company) means Dassault Systèmes, a French "société européenne" or any entity which Dassault Systèmes, directly or indirectly, (i) owns more than fifty percent (50%) of the outstanding equity or ownership interest, or (ii) has the power to designate the managing authority.

3DS Offering (or DS Offering) means one or more Licensed Programs and/or Online Services and/or Packaged Offerings.

3DS Website Terms means the relevant OST, Support Services policies, product portfolio, price, Data Processing Exhibit Service Level Agreement and shared responsibility model located at the website available at www.3ds.com/terms.

Administrative Region means either a country and, if applicable, an administrative territory with fax authority (e.g. rovinces, states and landers) on withholding tax, value-added tax or similar charges.

Agreement means these general terms, the applicable Country Specific Terms, the Transaction Documents, and the terms contained in the 3DS Website Terms referenced herein.

Applicable Data Protection Legislation means any applicable data privacy law and all other regulations that may apply to the Processing of Personal Data provided by Customer.

Controlled Data means any information: (i) required for the development, production, assembly, operation, repair, test, maintenance, or modification of an item, in any tangible or intangible form, (including but not limited to written or oral communications, designs, computer-aided design files, plans, models, photographs, instructions, engineering designs and specifications) falling in any category of the list of controlled items of any Export ControlRegulations; and/or (ii) subject to cybersecurity regulatory requirements on information technology systems used for its storage or transfer.

Controller, Data Subject, Personal Data, Personal Data Breach, Process/Processing, Processor, and Supervisory Authority when used in the appropriate context, shall have the same meaning as in the Applicable Data Protection Legislation. If the preceding terms are not defined in the Applicable Data Protection Legislation, then such terms as used in this Agreement shall have the same meaning as their analogous terms under the Applicable Data Protection Legislation. In the event there are no such analogous terms, then the definitions of those terms under Regulation (EU) 2016/679 (General Data Protection Regulation) shall apply.

Country Specific Terms means terms below corresponding to the country in which Customer is located, as such country is identified in the Transaction Document.

Customer Data means the data stored in the Online Services, including digital asset, metadata and Personal Data, which is either provided or generated by Customer, through its use of the Online Services.

Data Processing Exhibit means the terms published in the 3DS Website Terms applicable to the Processing of Personal Data by 3DS when 3DS is acting as Processor on behalf of Customer.

Distributor means a third party authorized by 3DS to distribute 3DS Offerings and/or Support Services.

Documentation means the current user documentation in any form or media as delivered together with the 3DS Offering for use in connection with the 3DS Offering.

Effective Date means (i) for a Licensed Program, the later of the following (a) the date on which such Licensed Program is shipped or made available electronically to Customer or, if applicable, (b) the date on which Customer is informed by 3DS that the associated license key can be requested or is available, or (ii) for Online Services, the date of delivery of the Online Services as described in Section 4.1.

Export Control Regulations means all applicable export control laws and regulations, and sanctions programs worldwide, including but not limited to the United States Export Administration Regulations (EAR) and the Office of Foreign Assets Control (OFAC) Specially Designated Nationals sanctions, and the European Union Council Regulation (EU) No 2021/821.

Licensed Program means (i) any data processing program and content, including databases and digital models, for which a license is provided to Customer pursuant to a Transaction Document, (ii) associated Documentation, and (iii) corrective patches and Releases to which a Customer is entitled. A Licensed Program does not include new versions of a Licensed Program, including any successor product which significantly differs in architecture, user interface, or mode of delivery.

Online Services means online access to, and use of, Licensed Program and/or other related services, as may be updated by 3DS from time to time and ordered by Customer pursuant to a Transaction Document. Online Services may also include certain Licensed Programs for which on-premises installation may be required.

OST (Offering Specific Terms) means specific terms applicable to a given Release of a 3DS Offering and published in the 3DS Website Terms. The applicable OST for a specific Release will be the one applicable for the specific Release at the time of the initial delivery. Any OST applicable to a subsequent Release of a 3DS Offering will not, on an overall basis, result in a material increase in Customer's liabilities and obligations or a material decrease in 3DS's liabilities and obligations under this Agreement.

Packaged Offering means a 3DS Offering composed of multiple Licensed Programs and/or Online Services as defined in the product portfolio available in the 3DS Website Terms, and ordered by Customer pursuant to a Transaction Document, provided that each Licensed Program or Online Service comprising a Packaged Offering shall be governed by all of the terms and conditions of this Agreement applicable to it respectively.

Release means a periodic update of the same major version of a 3DS Offering if and when made generally available to the market.

Service Level Agreement means the service level terms for the Online Services published in the 3DS Website Terms.

Sub-Processor means any Processor appointed by 3DS or by any other Sub-Processor of 3DS that receives, from 3DS or from any other Sub-Processor of 3DS, Personal Data for the Processing activities to be carried out on behalf of Customer in accordance with the terms of this Agreement and the terms of a written subcontract if applicable.

Support Services means the maintenance, enhancement and other support services for 3DS Offerings referred to herein and described in the 3DS Website Terms.

Transaction Document means the form provided by 3DS (which may be online) referencing this Agreement, signed or otherwise accepted by Customer, and accepted by 3DS, which identifies the 3DS Offerings and/or Support Services ordered by Customer, and includes other information such as the quantities of licenses thereof, fees payable (unless ordered through a Distributor), duration, geographical scope, and the 3DS Group Company serving as the licensor or service provider and Customer information.

Whenever required by the context hereof, the terms herein apply equally to the singular and plural forms of the terms and expressions defined.

2. License and Use Rights

- **2.1. Grant.** 3DS grants Customer, from the Effective Date, a non-exclusive and non-transferable (except as expressly permitted herein) right, for the duration identified in the Transaction Document, and solely for its own internal business use, to:
 - Access and use the 3DS Offering according to the terms and conditions of this Agreement and their applicable Documentation;
 - Allow its authorized users (as specified in the applicable OST) to access and use the 3DS Offering; and
 - Install the permitted number of copies and make one copy for back-up purposes for 3DS Offering requiring on-premises installation.
- **2.2. Scope.** Customer agrees to operate each 3DS Offering in accordance with the terms and provisions of this Agreement and applicable Documentation and to ensure that its authorized users comply with such terms and provisions. License keys, license tokens, or delivery of media do not in and of themselves, grant any right to use any software if such software has not been ordered pursuant to a Transaction Document. Except as expressly set forth in this Agreement, no other express or implied right or license is granted to Customer.

Except as specifically permitted in this Agreement or in a separate agreement, Customer agrees not to: (a) use and permit any third-party to use, any 3DS Offering and/or any part or output of any 3DS Offering, including content, data or results generated by the use of any 3DS Offering to develop software applications (i) for use by or distribution to any third party or (ii) competing or intending to compete directly or indirectly with any 3DS software or service, including through artificial intelligence model or system training; (b) rent, lease, or sublicense any 3DS Offering to third parties, or perform or offer any type of services to third parties relating to any 3DS Offering including, without limitation, consulting, training, assistance, outsourcing, service bureau, customization, or development; (c) correct errors, defects, and/or other operating anomalies of any 3DS Offering; (d) reverse engineer, decompile, disassemble, adapt, or otherwise translate all or part of any 3DS Offering; (e) conduct or authorize any third party to conduct security testing on Online Services or any related network or system used by 3DS to provide Online Services or Support Services; and (f) provide, disclose, or transmit publicly or to any third party any result of benchmarking or testing related to any 3DS Offering hosted by Customer or by a third party on its behalf, such as any vulnerability test result.

3. Support Services

Support Services for 3DS Offerings are provided in accordance with the applicable Support Services policy posted in the 3DS Website Terms and include support request management and, in the case of Licensed Programs, Releases made available during the Support Services term. Support Services policies are subject to change; but any change shall not take effect for Customer until Customer's next renewal of Support Services. Unless otherwise requested by Customer or terminated in accordance with this Agreement, Support Services shall automatically renew on an annual basis subject to the then-current Support Services policies.

4. Delivery and Payment

4.1. Delivery. 3DS Offerings will be delivered and/or made available to Customer electronically. Electronic delivery will be made by providing Customer with the necessary information to access the Online Services and/or download the Licensed Programs. Customer is responsible for accessing 3DS's website and downloading the Licensed Programs. Licensed Programs requiring physical delivery will be delivered by 3DS FCA (Incoterms 2020) 3DS's premises as designated by 3DS.

4.2. Payment

4.2.1. Payment Terms. Customer shall pay the fees applicable to each 3DS Offering and Support Services at the price identified in the applicable Transaction Document (or similar document between Customer and Distributor, if ordered through a Distributor). Unless otherwise agreed to in a Transaction Document, (i) all fees will be invoiced in advance, and (ii) Customer shall pay invoices in accordance with the Country Specific Terms. 3DS shall be entitled to suspend the provision of any Support Services and/or Online Services for which payment has not been made.

To avoid Customer having multiple renewal dates, the parties may elect to set a common renewal date with respect to any 3DS Offering or Support Services with different renewal dates, and 3DS will prorate the fees due as applicable so there is no duplication of fees.

Customer may elect to submit payment for certain 3DS Offerings online. In such a case, Customer will be notified by e-mail of upcoming renewals for those 3DS Offerings that are subject to automatic renewal. Renewal fees will be debited from Customer's account no earlier than seven (7) days prior to the renewal date. Once debited, the renewal shall be deemed accepted by Customer and cannot be canceled or withdrawn. Should such debit be rejected, 3DS shall be entitled to terminate the applicable 3DS Offering as of the renewal date.

- **4.2.2. Renewal Pricing.** The price for renewal of fixed term Licensed Program and/or Support Services for any given period is the price of the previous period, plus the last percentage of increase applicable to the Licensed Program in the applicable country, as published at least ninety (90) days prior to the renewal date. If applicable, current price increase terms and conditions can be found in the 3DS Website Terms. The price for renewal of fixed term Online Services and/or associated Support Services shall be calculated by applying the percentage difference between the list price of the renewal period and the list price of the prior period to the fees charged to Customer for the prior period.
- **4.2.3.** Late Payments. Customer shall pay interest and collection fees on late payments at the rate identified in the Country Specific Terms.
- **4.2.4.** Pay Agents. Customer may designate a third-party pay agent to be the recipient of 3DS invoices related to the 3DS Offerings and Support Services ordered under a Transaction Document, provided that such third-party pay agent is located in the same Administrative Region as the Customer, otherwise a separate agreement is needed. Such pay agent will make the applicable payments as specified in the Transaction Document directly to 3DS acting in the name of and on behalf of Customer. Customer shall remain liable to 3DS for all payment and tax obligations hereunder. Customer shall perform any obligation which could be created by or related to designation of a pay agent. No express or implied license to the 3DS Offerings or other benefit is granted to the third-party pay agent hereunder.
- **4.2.5. Taxes.** All prices are exclusive of taxes. Customer shall be responsible for payment of any and all taxes as more fully identified in the Country Specific Terms, and for any and all taxes due in relation to the transfer or usage of a 3DS Offering.

5. Intellectual Property

- **5.1. Ownership.** 3DS and/or its suppliers retain ownership in all intellectual property rights in all 3DS Offerings and all modifications or other derivative works thereof. Licensed Programs are licensed, not sold. Customer shall preserve and reproduce all copyright, patent, and trademark notices which appear in any 3DS Offering or any copy thereof. Customer recognizes that the methodologies and techniques contained in or expressed within the 3DS Offering are proprietary information or trade secrets of 3DS and/or its suppliers, whether or not marked as "confidential." Customer shall treat such as confidential information and not disclose them.
- **5.2. Intellectual Property Claims.** 3DS will defend Customer against any claim made by a third party that a 3DS Offering delivered under this Agreement infringes a copyright in any country or a patent of the United States, Japan, or a member state of the European Patent Organization. 3DS will also pay all costs, damages, and expenses (including reasonable legal fees) finally awarded against Customer by a court of competent jurisdiction or agreed to in a written settlement agreement signed by 3DS arising out of such claim. 3DS's obligations in this Section 5.2 are conditioned upon Customer (i) providing 3DS with prompt written notice of the claim, (ii) giving 3DS sole control of the defense of the claim and any related settlement discussion, and (iii) providing reasonable cooperation in the defense and settlement of the claim.

This obligation shall not apply to any claim related to (i) any modification of a 3DS Offering by anyone other than 3DS, (ii) Customer or third party content, including databases and/or digital models provided or published via the 3DS Offering, (iii) the use of one or more 3DS Offering in combination with other hardware, data, or programs not specified by 3DS, or (iv) the use of corrective patches or Releases other than the most recent one, to the extent that such claim would have been prevented by use of such patches or Releases.

If any claim of infringement of intellectual property rights is made, or in 3DS's reasonable opinion is likely to be made, 3DS may, at its sole discretion and expense either (i) secure the right for Customer to continue using the applicable 3DS Offering, (ii) modify it so that it is not infringing, or (iii) replace it with another program which is functionally comparable. If none of the foregoing options is available on terms which are reasonable in 3DS's judgment,

3DS may terminate the affected 3DS Offering. For fixed term 3DS Offerings, 3DS shall refund all prepaid but unused fees paid hereunder for the affected 3DS Offering. For other than fixed term 3DS Offering, 3DS shall either refund, or provide a credit to Customer, at Customer's option, in an amount equal to the corresponding one-time fee paid for the 3DS Offerings by Customer, depreciated on a straight-line over three (3) years. Prior to any such refund or credit, Customer shall certify, in writing signed by an officer of Customer, the return or destruction of all copies of the affected Licensed Program.

This Section 5.2 states 3DS's entire liability and Customer's exclusive remedy for any infringement of intellectual property rights.

6. Warranty

- **6.1. Licensed Program Warranty.** 3DS warrants for ninety (90) days from the initial delivery of each Licensed Program that such Licensed Program will materially conform to its Documentation when used in the specified operating environment. If the Licensed Program does not so conform, and Customer has notified 3DS within this warranty period, 3DS will use commercially reasonable efforts to make it conform as warranted. If 3DS has not corrected the non-conformity within ninety (90) days from the date of such notification, Customer may terminate the license to the non-conforming Licensed Program within thirty (30) days and receive a full refund of all fees paid for such non-conforming Licensed Program. This refund represents 3DS's sole liability and Customer's sole remedy for breach of this warranty.
- 6.2. Online Services Warranty. 3DS warrants, for the subscription term of each Online Services offering, that such Online Services will perform materially in accordance with its Documentation when accessed and used in accordance with terms and conditions of this Agreement. Customer shall provide 3DS with prompt notice of any claim under the warranty set forth above and, if applicable, provide 3DS with reasonable assistance required for 3DS to identify and repair such material non-conformance of the Online Services. Customer's sole and exclusive remedy for a breach of this warranty shall be that 3DS shall be required to use commercially reasonable efforts to make conform as warranted any non-conforming Online Services. If 3DS has not corrected the non-conformity within ninety (90) days from the date of such notification, Customer may terminate its right to access and use the non-conforming Online Services by providing written notice to 3DS within thirty (30) days and receive a pro-rata refund of prepaid subscription fees paid for the non-conforming Online Services covering the remainder of the term of such non-conforming Online Services after the effective date of termination. This refund represents 3DS's sole liability and Customer's sole remedy for breach of this warranty.
- **6.3. Disclaimers.** THE FOREGOING WARRANTIES ARE IN LIEU OF AND EXCLUDE ALL OTHER WARRANTIES, REPRESENTATIONS, OR CONDITIONS FOR 3DS OFFERINGS, WHETHER EXPRESS OR IMPLIED, INCLUDING, WITHOUT LIMITATION, ANY IMPLIED WARRANTY OF MERCHANTABILITY, FITNESS FOR A PARTICULAR PURPOSE, TITLE, AND NON-INFRINGEMENT.

3DS DISCLAIMS ALL LIABILITY FOR ANY USE OR APPLICATION OF ANY 3DS OFFERING OR THE RESULTS OR DECISIONS MADE OR OBTAINED BY USERS OF THE 3DS OFFERING. 3DS DOES NOT WARRANT THAT (I) THE FUNCTIONS OF ANY 3DS OFFERING WILL MEET CUSTOMER'S REQUIREMENTS OR WILL ENABLE IT TO ATTAIN THE OBJECTIVES CUSTOMER HAS SET FOR ITSELF, (II) THE 3DS OFFERING WILL OPERATE IN THE COMBINATION OR ENVIRONMENT SELECTED FOR USE BY CUSTOMER, OR (III) THE OPERATION OF THE 3DS OFFERING WILL BE UNINTERRUPTED OR FREE OF ERRORS. IN ALL INSTANCES, CUSTOMER SHALL BE SOLELY RESPONSIBLE FOR ENSURING THAT THE RESULTS PRODUCED BY THE 3DS OFFERING COMPLY WITH QUALITY AND SAFETY REQUIREMENTS OF CUSTOMER'S PRODUCTS OR SERVICES. NO EMPLOYEE OR AGENT OF 3DS IS AUTHORIZED TO GIVE A GREATER OR DIFFERENT WARRANTY. CUSTOMER SHALL HAVE EXCLUSIVE RESPONSIBILITY FOR (A) SELECTION OF THE 3DS OFFERING TO ACHIEVE CUSTOMER'S INTENDED RESULTS, (B) INSTALLATION OF THE LICENSED PROGRAM, (C) TAKING ADEQUATE MEASURES TO PROPERLY TEST, OPERATE, AND USE EACH 3DS OFFERING, AND (D) RESULTS OBTAINED THEREFROM.

3DS exercises no control over, and assumes no responsibility or liability for any Customer content or third party content, including databases and digital models provided or published via the 3DS Offerings.

7. Limitation of Liability

EXCEPT FOR 3DS'S LIABILITY UNDER SECTION 5.2, 3DS'S MAXIMUM LIABILITY FOR DAMAGES SHALL NOT EXCEED THE AMOUNT OF FEES ACTUALLY PAID BY CUSTOMER UNDER THIS AGREEMENT IN THE TWELVE (12) MONTHS PRIOR TO THE OCCURRENCE OF THE FIRST EVENT GIVING RISE TO A CLAIM HEREUNDER. THE EXISTENCE OF MULTIPLE CLAIMS SHALL NOT ENLARGE THIS MAXIMUM AND AGGREGATE LIMIT.

NOTWITHSTANDING THE FOREGOING, AND EXCLUDING 3DS'S OBLIGATIONS UNDER SECTION 5.2, TO THE EXTENT PERMITTED BY APPLICABLE LAW, 3DS SHALL HAVE NO LIABILITY FOR INDIRECT, INCIDENTAL, CONSEQUENTIAL, OR PUNITIVE DAMAGES, AS WELL AS CLAIMS FOR LOST PROFITS, BUSINESS INTERRUPTION, OR LOSS OF DATA, THAT IN ANY WAY RELATE TO THIS AGREEMENT, ANY 3DS OFFERING, DOCUMENTATION, OR SERVICES, WHETHER OR NOT 3DS HAS BEEN ADVISED OF THE POSSIBILITY OF SUCH DAMAGES.

THE LIMITATIONS STATED IN THIS SECTION 7 SHALL APPLY REGARDLESS OF THE FORM OF ACTION, WHETHER IN CONTRACT (INCLUDING WITHOUT LIMITATION, BREACH OF WARRANTY), TORT (INCLUDING WITHOUT LIMITATION, NEGLIGENCE), STATUTE, OR ANY OTHER LEGAL OR EQUITABLE REMEDY.

Customer agrees that any and all claims related to this Agreement shall be brought solely against 3DS and not any 3DS licensor or any other 3DS Group Company.

Any legal action against 3DS must be filed in the appropriate judicial jurisdiction within two (2) years after the applicable cause of action has arisen.

8. Distributors

Notwithstanding anything to the contrary herein, specifically including the terms of Section 4, for any 3DS Offering which Customer obtains through a Distributor, Customer agrees that Distributor is responsible for any order it accepts of (i) pricing, invoicing, payment collection, (ii) the delivery of 3DS Offering, (iii) the provision of Support Services as specified in the applicable Support Services policy and (iv) the transmission to 3DS of Customer's termination notification according to Section 9.3. Distributors are not agents of 3DS and have no ability to bind 3DS. 3DS remains independent from the Distributor and is not responsible for the Distributor's actions or omissions.

9. Term and Termination

9.1. Term. This Agreement remains in effect until terminated as provided hereunder, or until the expiration of all 3DS Offerings ordered under this Agreement.

9.2. Termination for Cause

- **9.2.1. Material Breach.** Either party may terminate this Agreement and/or Customer's rights to any or all 3DS Offerings and/or Support Services, if the other party is in material breach of any of its obligation, and fails to remedy such breach within thirty (30) days of receipt of written notice.
- **9.2.2.** Online Services Service Level Agreement Breach. Customer may terminate Online Services if 3DS fails to provide such Online Services in accordance with the then-applicable Service Level Agreement and fails to remedy such breach within thirty (30) days of receipt of written notice detailing such breach.

9.3. Termination for Convenience

- **9.3.1.** Licensed Program; and related Support Services. Customer may terminate the license to any Licensed Program by providing 3DS with at least thirty (30) days prior written notice. For perpetual licenses, such termination will take effect thirty (30) days after receipt of such notice; for fixed term licenses, such termination will take effect at the expiration of the then-current term. Absent notice from Customer in accordance with this Section 9.3.1 and unless otherwise stated in the applicable OST, fixed term licenses shall automatically renew. Customer may terminate Support Services for a Licensed Program by providing 3DS with at least thirty (30) days prior written notice, provided, however, that any such termination of Support Services shall apply to Support Services for all licenses of the relevant Licensed Program held by Customer under any agreement then in effect between Customer and any 3DS Group Company.
- **9.3.2.** Online Services; Use Rights and Support Services for Online Services. If applicable Customer may terminate the use rights and Support Services for Online Services, and either Customer or 3DS may terminate any Online Services, by providing the other party with thirty (30) days prior written notice. Absent such notification and unless otherwise stated in the applicable OST, the Online Services shall automatically renew and be subject to the then-applicable Service Level Agreement. In the case where Customer is terminating the use rights and Support Services related to Online Services, such termination shall take effect at the end of the notice period, in all other instances, such termination shall take effect at the next scheduled renewal date.

9.4. Effect of Expiration or Termination

9.4.1. General Effect. Upon expiration or termination of (i) this Agreement, (ii) a 3DS Offering, (iii) Support Services, (iv) Online Services, or (v) use rights and Support Services for Online Services, Customer shall not be entitled to receive any refund or credit for early cancellation or termination of any 3DS Offering or Support Services. Customer will have no further obligation to pay future fees associated with the terminated item(s), provided, however, that Customer shall continue to be obligated to pay all fees that have accrued or are otherwise owed by Customer hereunder. Customer shall no longer have access to the Online Services or Support Services. In the case of (i) or (ii), Customer shall immediately destroy or return all copies of the terminated or expired Licensed Program and associated Documentation. In the case of (iii), Customer shall only be authorized to use (subject to the terms of this Agreement) the latest Release deployed in production by Customer of the Licensed Program for which Support Services are expired or terminated. If Customer terminates Online Services in accordance with Section 9.2.2 herein, Customer shall receive a refund of any prepaid but unused recurring fees as of the termination date. Such refund represents 3DS's sole liability and Customer's sole remedy for 3DS' failure to provide Online Services in accordance with the Service Level Agreement.

- **9.4.2.** Agreement. In the event this Agreement is terminated by 3DS for breach, 3DS reserves the right to terminate (i) Customer's right to use any 3DS Offering for which payment has not been made, and (ii) Customer's right to use any or all 3DS Offering ordered hereunder if such breach is for other than a failure to pay.
- **9.4.3.** Support Services for Licensed Programs. Upon expiration or termination of Support Services for Licensed Programs, 3DS shall have no further obligation to provide any service or deliver any Release in support of any such license, except for providing license keys if necessary. Customer may reinstate Support Services, if available, provided such reinstatement is for all licenses of a given Licensed Program held by Customer under any license then in force between Customer and any 3DS Group Company, and Customer pays all fees that would have been due in respect of Support Services from the date of termination to the date of reinstatement of such Support Services, plus a reinstatement fee as set forth in the 3DS Website Terms.
- **9.4.4. Use Rights and Support Services for Online Services.** 3DS shall have no further obligation to provide any service related to expired or terminated Online Services, except as specified in the applicable OST. If permitted in the applicable OST, Customer may reinstate use rights and Support Services, if available, for Online Services, subject to payment of all fees that would have been due in respect of the use rights and Support Services for such Online Services from the date of termination to the date of reinstatement.

10. Customer Data; Data Privacy

- 10.1. Customer Data. Customer or its licensors remain the owner of Customer Data. Customer shall have the sole responsibility for the accuracy, quality, integrity, legality, reliability, suitability of and obtaining all necessary rights for all Customer Data. Customer grants to 3DS, 3DS Group Companies, and 3DS's subcontractors to the extent reasonably necessary, a non-exclusive, worldwide, royalty-free license to use, copy, store and transmit Customer Data to provide, maintain and adapt any software and services of 3DS or any 3DS Group Company. Customer shall defend the 3DS Group Companies against all third-party claims arising from or relating to (i) Customer's use of the Online Services in violation of applicable laws or regulations, and/or (ii) any violation, infringement, or misappropriation of the rights of a third party resulting from the Customer Data. Customer shall pay all costs, damages and expenses (including reasonable attorney's fees) finally awarded against 3DS by a court of competent jurisdiction or agreed to in a written settlement agreement signed by Customer arising out of any such claim, provided that (i) 3DS provides Customer with prompt written notice of the claim, and (ii) 3DS gives Customer sole control of the defense of the claim and any related settlement discussion and provides reasonable cooperation in the defense and settlement of the claim.
- 10.2. Customer Data: Storage. As part of the Online Services, and if available in the applicable OST, 3DS will provide storage of Customer Data for the duration of the Online Services and within the storage size limits defined in the applicable OST. Should Customer exceed such storage limits, Customer shall within fifteen (15) days of notice from 3DS, either (i) order additional storage capacity or (ii) reduce the size of the stored Customer Data accordingly.
- **10.3. Customer Data: Security.** For the security, integrity, availability and confidentiality of Customer Data stored in the Online Services, 3DS utilizes administrative, physical and technical safeguards aligned with accepted security industry standards, such as NIST SP 800-53 or ISO/IEC 27001. 3DS verifies such safeguards as described in 3DS trust center located at www.3ds.com/trust.

When using the Online Services, Customer is responsible for not compromising the security, integrity, availability and confidentiality of Customer Data, its network or systems and 3DS network or system by implementing measures aligned with applicable security industry standard, such as account authentication, update, backup and encryption measures as specified in the shared responsibility model available in the 3DS Website Terms.

10.4. Customer Data: Confidentiality. Customer Data shall be treated as confidential using no less than the same degree of care that 3DS uses with respect to its own confidential information of a similar nature, during the period when the Online Services are provided by 3DS and for a period of one (1) year following termination or expiration of Online Services.

To the extent necessary to perform this Agreement, 3DS is authorized to disclose Customer Data to third parties who have entered into an appropriate agreement with 3DS to protect the security and confidentiality of such Customer Data.

The obligation to treat Customer Data as confidential shall not apply to any Customer Data that is: (i) already in the possession of 3DS without any obligation of confidentiality at the time the information was received from Customer; (ii) independently developed by 3DS without reference to the Customer Data; (iii) or becomes publicly available without breach of this Agreement; (iv) rightfully received by 3DS from a third party without an obligation of confidentiality; or (v) required to be disclosed in accordance with a judicial or administrative decision but solely for disclosure to such judicial or administrative entity, provided that 3DS provides prompt information to Customer and reasonably cooperates with the Customer to limit the disclosure and use of the applicable information.

10.5. Personal Data: Controller Obligations. Each Party acknowledges that, for the Processing of professional contact details required for the management of this Agreement, Customer and 3DS each act as independent Controllers according to their own publicly available privacy policy. Customer acknowledges and agrees that it is and shall at all times remain the Controller of any Personal Data that will be Processed on its behalf as part of its access to and use of the Online Services.

Customer shall be responsible for complying with all Applicable Data Protection Legislation regarding, without limitation, (i) transfer of Personal Data, (ii) information of Data Subjects, and (iii) access, modification, and deletion rights of Data Subjects.

- **10.6. Personal Data: Location of Data Processing.** In order for 3DS to provide the Online Services, Customer appoints 3DS as Processor and agrees that Personal Data provided by Customer may be transferred to, stored, accessed, and Processed in any country in which 3DS or its Sub-Processor are located. 3DS will ensure that equivalent data protection obligations as set forth in this Agreement will be imposed on its Sub-Processor by way of a written agreement and/or the standard contractual clauses from the European Commission, as applicable, in such a manner that the Processing will address the requirements of the Applicable Data Protection Legislation.
- **10.7. Personal Data: Processor Obligations.** 3DS, acting as Processor, will, collect, store, and Process the Personal Data in accordance with this Agreement, including the Data Processing Exhibit published in the 3DS Website Terms.

If and as required by Applicable Data Protection Legislation, 3DS will:

- (i) for the duration of use of the Online Services, Process Personal Data provided by Customer in accordance with this Agreement and Customer's reasonable written instructions, which in all circumstances shall be consistent with this Agreement;
- require that the persons who are authorized to Process Personal Data provided by Customer have a need-to-know and are bound by an appropriate obligation of confidentiality;
- (iii) reasonably assist Customer in complying with its obligations as Controller, taking into account the nature of Processing as described in the Agreement;
- (iv) reasonably assist Customer with fulfilling its obligations with regards to responding to requests from Data Subjects to exercise their rights under Applicable Data Protection Legislation in a manner consistent with the use of the Online Services and 3DS's role as Processor;
- (v) make available to Customer the necessary information in 3DS's possession to demonstrate Customer's compliance with its obligations provided for in the Applicable Data Protection Legislation and reflected in this Section 10.7; and
- (vi) upon termination or expiration of the Agreement, delete or return r to Customer, at Customer's option, Personal Data part of Customer Data and delete all existing copies, in accordance with the terms of this Agreement, except where applicable law requires retention of Personal Data or where retention of Personal Data is necessary for evidentiary purposes during the applicable statute of limitations.

11. Additional Terms for Education Use and Non-Profit Research Use

- 11.1. Definitions Specific to Education Use and Non-Profit Research Use
- **11.1.1. Education Institution** means a Customer which is either (i) an accredited institution of education and/or research which is authorized to grant academic degrees (diploma or certificate) at any primary, secondary, or higher education level and/or (ii) an accredited institution of education delivering continuing education programs (e.g. training organization, training center, career school, vocational school).
- 11.1.2 Education Use means use of 3DS Offerings by authorized users at Education Institutions solely for purposes that are strictly related to (i) education, instruction, training, upskilling of duly enrolled participant in Customer's education program, (ii) research which is experimental and/or theoretical in nature, undertaken primarily to acquire new knowledge of the underlying foundations of phenomena and observable facts, up to proof of concept in a laboratory and/or (iii) experience-based and/or project-based learning activities which are under the direction and supervision of an Education Institution's faculty, undertaken to perform a project sponsored by one or more third parties to address problems presented by such third parties.
- **11.1.3. Non-Profit Research Institutes** means a Customer which is a non-profit organization dedicated to research, notwithstanding anything to the contrary in any Country Specific Term.
- **11.1.4. Non-Profit Research Use** means use of 3DS Offerings by authorized users at Non-Profit Research Institutes solely for purposes related to research that is experimental and/or theoretical in nature, undertaken primarily to acquire new knowledge of the underlying foundations of phenomena and observable facts, up to proof of concept in a laboratory.
- **11.2. Education Use or Non-Profit Research Use Restrictions.** Notwithstanding anything to the contrary in Section 2 or any country-specific terms, 3DS Offerings for Education Use or Non-Profit Research Use shall not be used, directly or indirectly, for commercial purposes of Customer or any third party (e.g. for production of goods and services for resale, professional consulting, corporate or government internships, work-for-hire under commercial and/or governmental contract terms).
- **11.3. Content Watermarking.** Content produced using any 3DS Offering for Education Use may automatically contain a watermark identifying the 3DS Offering used. Customer shall not remove any such watermark.

12. Export

- **12.1. Export Regulations.** 3DS Offerings, Support Services and related Documentation are subject to Export Control Regulations concerning (i) their export, import, re-export and in-country transfer, (ii) their intended end use by Customer, and (iii) the receipt, use, storage and/or export of Controlled Data. In the event of any conflict between Export Control Regulations, for purposes of the Agreement, the most restrictive law shall prevail.
- **12.2. Licenses/Authorizations.** 3DS, any 3DS Group Company and their licensors shall not be liable if authorizations, licenses and approvals, required for export to Customer under Export Control Regulations, are not sought or obtained.
- 12.3. Customer Responsibilities. Customer shall comply with all applicable Export Control Regulations, and shall not export or re-export, directly or indirectly, any 3DS Offering or related Documentation if such export or re-export requires an export license or other governmental approval pursuant to Export Control Regulations without first obtaining such license or approval. Customer shall not use any 3DS Offering and Documentation in violation of Export Control Regulations, including but not limited to proliferation of any nuclear, chemical or biological weapons or missile delivery systems. Customer shall not divert any 3DS Offering or Documentation to any country, company or individual if prohibited by Export Control Regulations. For that purpose, 3DS shall be entitled at any time to obtain from the Customer that it (i) signs a binding end-use form confirming Customer's compliance with Export Control Regulations (in particular as regards the compliant use and absence of diversion of 3DS Offerings), and (ii) provides evidence to authenticate the use of 3DS Offerings in compliance with Export Control Regulations.
- **12.4. Online Services.** Customer recognizes that Customer's data may be transferred to, or stored in any country. In the event of export, Customer shall be deemed the exporter of Controlled Data and shall be responsible for compliance with all Export Control Regulations applicable to its data. Customer shall ensure that all users under the Agreement abstain from processing, storing or uploading any Controlled Data on its data sharing environment.
- **12.5. Applicability.** The export, re-export, and in-country transfer of information under the Agreement, with respect to the provision of Support Services and all other activities, is subject to compliance with Export Control Regulations. Unless provided for in a separate agreement, the parties shall not disclose or exchange any Controlled Data.
- 12.6. Compliance. The obligations of 3DS and any 3DS Group Company under the Agreement shall be subject to and conditioned upon compliance with all applicable Export Control Regulations. 3DS and/or any 3DS Group Company may terminate or suspend at any time the Agreement, all licenses and access to Online Services, provision of 3DS Offerings or Support Services hereunder, or generally cease or suspend the performance of its obligations under the Agreement, if Customer violates the Export Control Regulations provisions of the Agreement or if performance of the Agreement would cause 3DS and/or any 3DS Group Company to infringe any Export Control Regulations or to be potentially exposed to any sanctions or penalties that could be imposed by any governmental authority as a result of continued performance. Such suspension or termination shall be effective upon written notice as of the date stipulated in that notice.

13. Software Compliance

13.1. Unauthorized Use Detection. 3DS Group companies employ measures to eliminate unauthorized use of 3DS software. 3DS software may include a security mechanism that can detect the installation or use of illicit copies of 3DS software and that is able to collect and transmit data about such illicit copies only. Data collected by such mechanism will not include any data created by Customer with the 3DS software. Customer consents to such detection and collection of data, as well as its transmission to and use by 3DS.

Further, 3DS also reserves the right to use a hardware lock device, license administration software, and/or a license authorization key to control access to, and use of, any 3DS Offering. Customer shall not take steps to tamper with, circumvent, or disable any such measure.

Compliance Verification. Customer shall maintain at all times accurate and up to date records (with at least three (3) years history) related 13.2. to its use of the 3DS Offerings in compliance with the applicable agreement ("Records"). Such Records shall include, without limitation, information related to logs (including usage tracing logs), Machines and Users. Upon written notification by 3DS, Customer will promptly provide such Records to 3DS or to such third party as 3DS may appoint ("Auditor") in order to assess if Customer's use of 3DS software is compliant with the terms of a valid agreement ("Verification Purpose"). Within five (5) business days of 3DS issuing the notification, Customer shall (i) appoint a single point of contact, (ii) comply with the requests outlined in the notification, and (iii) participate in a kick-off meeting during which the assessment procedure will be defined. The document establishing this agreed procedure will be provided by 3DS or its Auditor following the kick-off meeting. Data collection shall begin within ten (10) business days from the date 3DS issued the notification. For the Verification Purpose, Customer agrees to cooperate fully and in a timely manner with 3DS and its Auditor. This includes participation in regular meetings (during normal business hours), responding to questionnaires and information requests, and during normal business hours, providing access to (i) Customer's premises or any premises where 3DS Offerings are installed, (ii) machines, and (iii) system tools outputs, such as active directories or software asset management (SAM) tools. Customer shall also allow the execution of all appropriate tools provided by 3DS or its Auditor to generate audit records. For the Verification Purpose, notwithstanding anything to the contrary in this Agreement and in case the audited 3DS Offering is Online Services, 3DS or its Auditor shall have access to Customer Data related solely to Customer's use of or access to Online Services. In this case, 3DS shall first notify Customer of this audit, specifying the type of Customer Data to be accessed to by 3DS and its Auditors. Customer data collected by 3DS and its Auditor pursuant to this Section 13.2 shall be used solely by 3DS and its Auditor for the Verification Purpose, and shall not be disclosed to any other third party or used for any other purpose. If the assessment reveals unauthorized use of any 3DS Offerings, Customer shall promptly and not later than thirty (30) days after receiving the final assessment report (i) cease any unauthorized use (ii) pay 3DS any amount owed as a result of such unauthorized use at the then-current list price, and (iii) reimburse 3DS for the cost of such assessment. Nothing in the above waives 3DS's rights to enforce the Agreement or to protect its intellectual property by any other means permitted by law or in equity. Each party's failure to comply with the provisions of this Section 13.2 shall constitute a material breach of this Agreement.

14. Miscellaneous

- 14.1. Notices. All notices required hereunder shall be in writing and delivered to the other party at its contact address as specified in the relevant Transaction Document. In the case of Customer providing a notice to 3DS, any notice shall include a copy to legal.notice@3ds.com. For Online Services, notices may also be delivered by 3DS by e-mail and deemed to have been given twenty-four (24) hours after the time such e-mail is sent to Customer. For 3DS Offerings ordered on the 3DS web store, notices shall be delivered pursuant to the processed defined on such 3DS web store. Notwithstanding the foregoing, where in Section 9.3 a written notice is required, that requirement can be met by exchange of letters, emails or other electronic means for which a processed has been provided by 3DS.
- **14.2.** Force Majeure. Neither party hereto shall be liable for any default in the performance of its obligations under this Agreement resulting from (i) a force majeure event as defined by the law governing this Agreement and the courts in such jurisdiction, or (ii) the following causes: strikes (whether previously announced or not), war (declared or not), riots, governmental action, acts of terrorism, acts of God (fire, flood, earthquake, etc.), or any electrical, utility, or telecommunication outage.
- 14.3. Third Party Hosting. Customer is authorized to install and use the Licensed Programs remotely on machines operated by a well-established third party service provider and to appoint such provider to host the Licensed Programs solely for and on behalf of Customer; provided, however, that (i) only duly authorized users shall have the right to use the Licensed Programs, (ii) Customer shall ensure that such service provider grants access to the Licensed Programs solely for the purpose of providing the services mentioned above to Customer in compliance with the terms of this Agreement, and (iii) such service provider is not part of a group of companies that provides products or services competing with 3DS Offerings. If Customer becomes aware of any unauthorized access, use, or disclosure of the Licensed Programs, Customer shall immediately terminate the service provider's access to the Licensed Programs and promptly notify 3DS or the relevant 3DS Group Company.
- **14.4. Severability.** If any provision of this Agreement is found to be illegal, void, or unenforceable, the other provisions shall remain in full force and effect, and the affected provision will be modified to render it enforceable and effective to the maximum extent possible in order to effect the original intent of the parties.
- 14.5. Transfer; Assignment; Subcontract. Any subcontract, assignment, delegation, or other transfer of this Agreement or any right, duty, benefit, or obligation of Customer hereunder (including without limitation by operation of law or by sale of Customer assets, whether directly or by merger, and a change in control of Customer shall be deemed to be an "assignment" for purposes of the foregoing) is subject to 3DS's prior written approval. Any attempt to do so without 3DS consent is void. Any approved transfer may be subject to an adjustment fee. This Agreement shall be binding upon, and inure to the benefit of 3DS and its successors and assigns. 3DS may assign, delegate, subcontract, or otherwise transfer any of its rights or obligations hereunder, in whole or in part, without Customer's consent.
- 14.6. Entire Agreement. This Agreement comprises the complete agreement between the parties relating to the subject matter hereof and supersedes all prior and contemporaneous proposals, agreements, understandings, representations, and communications, whether oral or written. Customer confirms that it (i) has full knowledge of all terms herein and those incorporated herein by reference, (ii) agrees to be bound by and to comply with such terms, and (iii) in entering into this Agreement, has not relied upon the future availability or functionality or product updates with respect to any 3DS Offering. Any term or condition contained and/or referenced in Customer's purchase order shall not in any way supersede, supplement, or otherwise modify the terms of this Agreement. No waiver or modification of any provision of this Agreement shall be binding unless made in writing and signed by both parties. A party's failure at any time to require performance of any provision hereof shall in no manner affect its right at a later time to enforce that or any other provision.
- **14.7. Order of Precedence.** If there is a discrepancy, inconsistency, or contradiction between any OST and these terms, the provisions of the corresponding OST shall prevail, but solely with respect to those 3DS Offerings described in such OST.
- **14.8. References.** Customer authorizes 3DS to use its name and logo on 3DS's marketing and communication materials, including its website and/or social media accounts, for the purposes of identifying Customer as a client of 3DS.
- **14.9. Language.** This Agreement is provided in English and may be provided, for informational purposes only, in a language other than English. The English version shall be the only binding and enforceable version of this Agreement.
- **14.10. Survival.** The following sections of this Agreement shall survive termination or expiration: Sections 1 ("Definitions"), 2.2 ("Scope"), 4 ("Delivery and Payment"), 5 ("Intellectual Property"), 6.3 ("Disclaimers"), 7 ("Limitation of Liability"), 8 ("Distributors"), 9.4 ("Effect of Expiration or Termination"), 10 ("Customer Data; Data Privacy"), 11 ("Additional Terms for Education Use and Non-Profit Research Use"), 12 ("Export"), 13 ("Software Compliance"), and 14 ("Miscellaneous") and any term identified as surviving termination in the Country Specific Terms.

14.11.	Governing Law and Jurisdiction.	The governing law and jurisdiction are set forth in the Country Specific Terms.

COUNTRY SPECIFIC TERMS

TERMS SPECIFIC TO EUROPEAN ECONOMIC AREA COUNTRIES

Customer Data; Data Privacy – This clause is applicable if at least one of the parties is located in EEA. In case of conflict between these
terms and another section that amends this section 10 applicable to Customer, such local section shall prevail.

The second paragraph of Section 10.4 is amended as follows:

To the extent necessary to perform this Agreement and to the maximum extent permitted by law, 3DS is authorized to disclose Customer Data to third parties who have entered into an appropriate agreement with 3DS to protect the security and confidentiality of such Customer Data.

The provisions of Section 10.7 are amended as follows:

3DS, acting as Processor, will, collect, store, and Process the Personal Data in accordance with this Agreement, including the Data Processing Exhibit published in the 3DS Website Terms.

If and as required by Applicable Data Protection Legislation, 3DS will:

- (i) for the duration of use of the Online Services, Process Personal Data provided by Customer in accordance with this Agreement and Customer's reasonable written instructions, which in all circumstances shall be consistent with this Agreement;
- (ii) require that the persons who are authorized to Process Personal Data provided by Customer have a need-to-know and are bound by an appropriate obligation of confidentiality;
- (iii) reasonably assist Customer in complying with its obligations as Controller, regarding sections 32 to 36 of the General Data Protection Regulation, taking into account the nature of Processing as described in the Agreement; If 3DS has reason to believe or is convinced that a Personal Data Breach impacting Customer has occurred, 3DS will (i) notify the incident to Customer without undue delay after becoming aware of such Personal Data Breach, (ii) provide Customer with available information allowing it to comply with its notification obligations with competent supervisory authority;
- (iv) reasonably assist Customer with fulfilling its obligations with regards to responding to requests from Data Subjects to exercise their rights under Applicable Data Protection Legislation in a manner consistent with the use of the Online Services and 3DS's role as Processor;
- (v) make available to Customer the necessary information in 3DS's possession to demonstrate Customer's compliance with its obligations provided for in the Applicable Data Protection Legislation and reflected in this section; and , in case compliance with Applicable Data Protection Legislation cannot be evidenced through the appropriate documentation provided by 3DS, allow for, an audit. Such audit will be (i) notified to 3DS in writing at least thirty (30) days in advance by indicating its scope which shall be limited to assess Customer's compliance where the documentation provided by 3DS is not relevant (ii) conducted by an independent auditor mandated by Customer at Customer's costs and performed not more than once every twelve (12) months;
- (vi) implement the technical and organization measures as described in the Data Processing Exhibit, which Customer deems appropriate taking into account the state of the art, the costs of implementation and the nature, scope, context and purposes of Processing as well as the risk of varying likelihood and severity for the rights and freedoms of natural persons.
- (vii) keep a list of the Sub-Processors that will be involved in the Processing of Customer's Personal Data due to the Processing activities implemented on behalf of Customer and inform Customer of any intended changes concerning the addition or replacement of other Sub-Processors, thereby giving the Customer the opportunity to object to such changes. Customer will be notified at least 15 (fifteen) days in advance before authorizing any new Sub-Processor to Process Customer's Personal Data with a mechanism to obtain notice of that update, except in case of emergency. Customer may reasonably object to 3DS's use of a new Sub-Processor if (i) such new Sub-Processor Processes Customer's Personal data, (ii) Customer demonstrates it has a legitimate interest, and notifies 3DS in writing, within fifteen (15) days after receipt of the notification, it being specified that in the absence of an objection from Customer, the Sub-Processor is deemed to be accepted by Customer. If Customer notifies its objection related to the new Sub-Processor within the above timeframe, Customer may terminate the 3DS Offering impacted by this change of Sub-Processor before the end of the notice period of fifteen (15) days after receipt of the notification;

(viii) upon termination or expiration of the Agreement, delete or return to Customer, at Customer's option, Personal Data part of Customer Data and delete all existing copies, in accordance with the terms of this Agreement, except where applicable law requires retention of Personal Data or where retention of Personal Data is necessary for evidentiary purposes during the applicable statute of limitations.

TERMS SPECIFIC TO EUROPEAN UNION COUNTRIES

The following Sections apply to Customers based in European Union countries.

Customer Data; Data Privacy

In addition to the provisions of Section 10.1 of the General Terms, the following shall apply:

Customer may request to limit the usage of Customer Data by 3DS to provide, maintain and optimize the Online Services by sending an email to legal.customer-data@3ds.com which must include the following information: the name and reference of this Agreement, Customer's full name and address, as well as the 3DS reference number of the Transaction Document of the concerned Online Services.

In addition to the provisions of Section 10 of the General Terms, the following shall apply:

10.8. Switching Process

10.8.1. Switching Process Definitions

For the sole purpose of this Section "Switching Process", the following definitions are added to Section "Definitions" of this Agreement:

Data Register means the information with relevant details related to Exportable Data, available at https://www.3ds.com/trust.

Exportable Data means Customer Data stored in the Online Services and subject to the Switching Process, as described in more details in the Data Register.

Switching Notice Period means the period of two (2) months for the initiation of the Switching Process from receipt of Customer's Switching Request.

Switching Procedures means information on available procedures for the Switching Process, published at https://support.3ds.com/knowledge-base/switching-procedures/

Switching Process means the process governed by and described in Section 10.8 herein, in accordance with the European Union Data Act (Regulation (EU) 2023/2854 of the European Parliament and of the Council of December 13th 2023).

Switching Request means Customer's request to 3DS, as described in the Switching Procedures, whereby Customer indicates to 3DS its intention to (i) change from using Online Services to using other data processing services covering the same service type offered by another data processing provider, (ii) change to an on-premises infrastructure operated by Customer or by a third party on its behalf, and/or (iii) delete of its Exportable Data.

Transitional Period means the thirty (30) days period starting from the day following the end of the Switching Notice Period, unless extended as set out in Sections 10.8.3.2 and 10.8.3.3.

Exportable Data Retrieval Period means the period of thirty (30) calendar days for the retrieval of Exportable Data, starting on the day following the end of the Transitional Period agreed between Customer and 3DS.

Switching Assistance Services means reasonable assistance provided by 3DS to Customer in the Switching Process during the Transitional Period, as described in the Switching Procedures.

10.8.2. Initiation of the Switching Process

Subject to the Switching Notice Period, Customer may initiate the Switching Process by sending a Switching Request in accordance with the Switching Procedures. Any Switching Request must be submitted by Customer at least ninety (90) calendar days before the date of expiration or termination of the Online Services.

10.8.3. Duration of the Switching Process

- **10.8.3.1. Transitional Period**. 3DS will prepare the retrieval of Exportable Data Retrieval Period. The Agreement, including the provision of the Online Services, will remain applicable during the Transitional Period and any extension thereof, as set out in Sections 10.8.3.2 and 10.8.3.3. Any such extension will be subject to additional fees.
- **10.8.3.2. Extension of the Transitional Period by 3DS**. If it is technically unfeasible to meet the initial deadline, 3DS can extend the duration of the Transitional Period, and in such case will:
- a) notify Customer within fourteen (14) working days of the reception of the Switching Request;
- b) document the technical unfeasibility; and
- c) indicate an alternative transitional period, which shall not exceed seven (7) months.
- **10.8.3.3. Extension of the Transitional Period by the Customer**. Customer has the right to extend the Transitional Period once for a period it considers more appropriate for their own purposes, which shall not exceed seven (7) months. In this case, Customer must inform 3DS in writing either within the Switching Request or within seventy-five (75) calendar days thereafter, and indicate the duration of the requested extended Transitional Period, in accordance with the Switching Procedures.
- **10.8.3.4.** Exportable Data Retrieval and Deletion Periods. On the day following the end of the Transitional Period, Customer has thirty (30) calendar days to retrieve the Exportable Data. At the end of the Exportable Data Retrieval Period and once the Switching Process has been successfully completed, 3DS undertakes to delete Customer Data within thirty (30) calendar days, and will notify Customer of such deletion.
- 10.8.4. Obligations of 3DS during the Switching Process. Throughout the Switching Process, 3DS undertakes to:
- (i) implement the Switching Process, in accordance with the Switching Procedures;
- (ii) provide Switching Assistance Services, in accordance with the Switching Procedures;
- (iii) act with due care to maintain business continuity in the provision of the Online Services under the Agreement;
- (iv) provide available information concerning risks to business continuity, known to 3DS, in the provision of the Online Services, in accordance with the Switching Procedures; and
- (v) maintain security of Customer Data, in accordance with this Agreement.
- 10.8.5. Obligations of the Customer during the Switching Process. Throughout the Switching Process, Customer undertakes to:
- (a) cooperate with 3DS in good faith, and ensure the cooperation of the destination services provider if any, to (i) make the Switching Process effective,
- (ii) enable the timely transfer of Exportable Data and (iii) maintain the business continuity of the Online Services, including payment of additional fees if any applicable;
- (b) take all reasonable steps to ensure the effectiveness of the Switching Process, including importing and implementing Exportable Data in its own systems or in the systems of the destination service provider; and
- (c) respect the intellectual property rights and trade secrets of 3DS, or its licensors, as part of the Switching Process.
- **10.8.6. Charges**. Any Switching Request from Customer which goes beyond the obligations of 3DS regarding the Switching Process stipulated in Section 10.8 of the Agreement, or which constitutes a highly complex or costly request or for which it is impossible to switch without significant interference in the Exportable Data or on the architecture of the Online Service, is considered to be additional services subject to their own terms and separate charges in accordance with the Switching Procedures.
- 10.8.7. End of the Switching Process. 3DS will notify the Customer of the end of Switching Process, including the end of the provision of associated Online Services specified by Customer in the Switching Request. If the Customer has chosen to switch and does not dispute the completion of the Switching Process within fourteen (14) calendars days after receiving such notification, the Switching Process shall be deemed successfully completed. If the Customer has chosen not to switch but to delete Exportable Data instead, the Switching Process shall be deemed successfully completed at the end of the Switching Notice Period.

Notwithstanding any provisions to the contrary in the Agreement, upon the end of the Switching Process, Customer shall pay to 3DS all remaining fees owed by Customer for the Online Services specified in the Switching Request for the entire periods to which the Customer has committed under this Agreement.

TERMS SPECIFIC TO EUROPE MIDDLE EAST AND AFRICA

Payment Terms – In addition to the provisions of Section 4.2.1 of the General Terms, the following shall apply:

Unless otherwise agreed to in writing by 3DS, Customer shall pay all invoices by wire transfer within thirty (30) days from the date of invoice.

Late Payments – In addition to the provisions of Section 4.2.2 of the General Terms, the following shall apply:

Customer shall pay interest for late payment at a rate of twelve (12) percent per year on all sums unpaid at the due date, plus an amount of forty (40) Euros, which shall be increased if the recovery costs incurred in collecting the unpaid sums exceed such amount.

Taxes – In addition to the provisions of Section 4.2.3 of the General Terms, the following shall apply:

All prices are exclusive of taxes. Customer shall be responsible for payment of any and all taxes, including fees, duties, excises, import VAT, or similar charges of any nature whatsoever, now in force or enacted in the future, that are levied, assessed, charged, withheld, or collected for or in connection with the transfer or usage, to the extent authorized hereunder, of the 3DS Offerings provided hereunder or otherwise arising in connection with this Agreement, but excluding domestic taxes based on 3DS's net income.

If Customer is required to withhold, deduct, or pay for any tax from the amount of fees to be paid under this Agreement, then Customer shall pay such additional amount to 3DS as is necessary to ensure that 3DS receives a sum equal to what would have been received had no such withholding, deduction or payment been required.

Interoperability – In addition to the provisions of Section 5.1 of the General Terms, the following shall apply:

Except to the extent permitted by applicable law, Customer shall not modify, adapt, reverse engineer, decompile, disassemble, or otherwise translate all or part of the 3DS Offerings. In the event Customer wishes to ensure the interoperability, within the limits of its authorized use as defined in Section 2 of these General Terms, of the 3DS Offerings with other computer software or with equipment under conditions provided for by law (including without limitation laws implementing the directive 2009/24/EC of the European Parliament and of the Council of 23 April 2009 on the legal protection of computer programs), Customer must ask 3DS for a license to use standard interfaces, exclusively for internal use to achieve interoperability. 3DS will grant Customer a license to use the standard interfaces at the then current prices and contractual conditions of 3DS or, if standard interfaces are not available, 3DS, for a fee, may provide Customer with the necessary information to permit interoperability. Customer is not authorized to give access to these interfaces to any person other than its authorized users (as defined in the OST).

Governing Law and Jurisdiction – In addition to the provisions of Section 14.11 of the General Terms, the following shall apply:

The Agreement shall be governed by and construed in accordance with the laws of France. All disputes arising out of or in connection with this Agreement shall be finally settled under the Rules of Arbitration of the International Chamber of Commerce by one or more arbitrators appointed in accordance with the said Rules of Arbitration. The seat of the arbitration shall be in Paris, France. The arbitral proceedings shall be conducted in English. Any party to this Agreement shall have the right to have recourse to and shall be bound by the Pre-Arbitral Referee Procedure of the International Chamber of Commerce in accordance with its Rules for a Pre-Arbitral Referee Procedure. Moreover, either party is entitled to seek at any time interim relief (including injunctive relief) before any State court of competent jurisdiction.

TERMS SPECIFIC TO AFRICAN COUNTRIES

Disclaimers – In addition to the provisions of Section 6.3 of the General Terms, the following shall apply:

3DS hereby specifically records that, due to the electronic nature of the transactions concluded for the licensing of any Licensed Program, 3DS has no knowledge of and has not received any information from Customer pertaining Customer's intended use or the purpose for which the license for the Licensed Program is acquired under this Agreement.

Governing Law and Jurisdiction – In addition to the provisions of Section 14.11 of the General Terms, the following shall apply:

This Agreement shall be governed and construed in accordance with the laws of the Republic of South Africa. The High Court of the Republic of South Africa shall have non-exclusive jurisdiction to hear any dispute arising out of or in connection with the interpretation and/or performance of this Agreement, however, the parties acknowledge and agree that in the event that the subject matter of any such dispute is Intellectual Property, 3DS shall have the right to bring any such dispute under the rules of the Arbitration Foundation of Southern Africa ("AFSA"), by one or more arbitrators appointed by the Secretariat of AFSA.

In such event, the seat of the arbitration shall be in Sandton, South Africa. The arbitral proceedings shall be conducted in English. The references to AFSA shall include its successor or body nominated in writing by it in its stead.

Customer acknowledges and agrees that the foregoing shall not prevent, restrict or otherwise limit in any manner, 3DS's rights to seek equitable remedies, including injunctive relief before any competent court in any jurisdiction.

This Section is a separate, divisible agreement from the rest of this Agreement and shall remain in effect even if this Agreement terminates, is nullified or cancelled for whatsoever reason or cause.

• Consumer Rights – In addition to the provisions of Section 14 of the General Terms, the following shall apply:

In the case where Customer is subject to the ambit of any applicable consumer protection laws or regulations or has any rights under any such consumer protection laws or regulations then, notwithstanding anything to the contrary contained in the Agreement or any other terms applicable to Customer, any provision in the Agreement which is directly inconsistent with or has the effect of excluding any inalienable right afforded to Customer by virtue of such consumer protection laws or regulations shall be null and void, subject to Section 14.4 of the General Terms.

TERMS SPECIFIC TO BOSNIA AND HERZEGOVINA

Transfer; Assignment; Subcontract – In addition to the provisions of Section 14.5 of the General Terms, the following shall apply:

For the avoidance of doubt, and subject to applicable law in Bosnia and Herzegovina, in case an approval or a consent or a licence for transfer, assignment or subcontract, including but not limited to the set off/payment netting/assignment of rights and obligations under this Agreement from the competent authorities in Bosnia and Herzegovina is required then the Customer shall undertake all required efforts to obtain such approval or consent so the transfer, assignment or subcontract may occur as stipulated in this Agreement.

TERMS SPECIFIC TO BULGARIA

Taxes – In addition to the provisions of Section 4.2.3 of the General Terms, the following shall apply:

If Customer is required to withhold, deduct, or pay for any tax from the amount of fees to be paid under this Agreement, including but not only withholding tax on royalties or payments for technical services, then Customer shall (i) pay the relevant taxes to the tax authorities within the terms under the applicable law and provide 3DS promptly with evidence for payment of such taxes; and (ii) assist upon request by 3DS with performing such actions and signing such documents as may be required for 3DS to benefit from any available double tax treaty relief. If Customer is required to withhold, deduct, or pay for any tax from the amount of fees to be paid under this Agreement, then Customer shall pay such additional amount to 3DS as is necessary to ensure that 3DS receives a sum equal to what would have been received had no such withholding, deduction or payment been required.

Limitation of Liability – In addition to the provisions of Section 7 of the General Terms, the following shall apply:

The limitations of liability set out in Section 7 of the General Terms apply to the maximum extent permitted by applicable law.

TERMS SPECIFIC TO HUNGARY

Limitation of Liability – The provisions of Section 7 of the General Terms shall be entirely deleted and replaced as follows:

EXCEPT FOR 3DS'S LIABILITY UNDER SECTION 5.2 HEREOF, 3DS'S MAXIMUM LIABILITY FOR DAMAGES SHALL NOT EXCEED THE AMOUNT OF FEES ACTUALLY PAID BY CUSTOMER FOR THE LICENSED PROGRAM OR THE ONLINE SERVICES WHICH CAUSED THE DAMAGES IN THE PRECEDING TWELVE (12) MONTH-PERIOD PRIOR TO THE OCCURRENCE OF THE CAUSE OF ACTION GIVING RISE TO THE CLAIM.

3DS SHALL HAVE NO LIABILITY FOR INDIRECT, INCIDENTAL, CONSEQUENTIAL OR PUNITIVE DAMAGES, INCLUDING WITHOUT LIMITATION CLAIMS FOR LOST PROFITS, BUSINESS INTERRUPTION AND LOSS OF DATA, THAT IN ANY WAY RELATE TO THIS AGREEMENT, ANY 3DS OFFERING, DOCUMENTATION OR SERVICES, WHETHER OR NOT 3DS HAS BEEN ADVISED OF THE POSSIBILITY OF SUCH DAMAGES AND NOTWITHSTANDING THE FAILURE OF THE ESSENTIAL PURPOSE OF ANY REMEDY.

THE LIMITATIONS STATED IN THIS SECTION SHALL APPLY REGARDLESS OF THE FORM OF ACTION, WHETHER THE ASSERTED LIABILITY OR DAMAGES ARE BASED ON CONTRACT (INCLUDING, BUT NOT LIMITED TO, BREACH OF WARRANTY), TORT (INCLUDING, BUT NOT LIMITED TO, NEGLIGENCE), STATUTE, OR ANY OTHER LEGAL OR EQUITABLE THEORY.

Customer waives any and all claims related to this Agreement or any 3DS Offerings or Documentation or services provided hereunder, for any direct, incidental or consequential damages, on any basis, against any 3DS licensors or any 3DS Group Company other than 3DS.

Any legal action against 3DS must be filed with the appropriate judicial jurisdiction within two (2) years after the applicable cause of action has arisen. Any legal action against 3DS for a damage or loss caused intentionally or by criminal conduct must be filed within five (5) years after the applicable cause of action has arisen, or if the given crime is punishable for a longer period of time, that longer period of time.

Miscellaneous – In addition to the provisions of Section 14 of the General Terms, the following shall apply:

Customer and 3DS mutually agree not to apply section 5 subsection 2 of Hungarian Act No CVIII of 2001.

TERMS SPECIFIC TO IRAQ

■ Late Payments – In addition to the provisions of Section 4.2.2 of the General Terms, the following shall apply:

Customer shall pay interest for late payment at a rate of seven (7) percent per year on all sums unpaid at the due date, and reasonable attorneys' fees and costs incurred in collecting unpaid amounts.

Export – Section 12 of the General Terms shall be deleted and replaced in its entirety with the following:

Export to Customer of 3DS Offering and Documentation is subject to all applicable countries' export and re-export laws and regulations. 3DS and its licensors shall have no liability towards Customer if necessary authorizations, licenses or approvals are not obtained. Customer shall not export or re-export, either directly or indirectly, 3DS Offering when such export or re-export requires an export license or other governmental approval without first obtaining such license or approval. Customer hereby warrants to 3DS that all 3DS Offerings ordered hereunder shall not be used in violation of any applicable export laws, including for proliferation of any nuclear, chemical or biological weapons or missile delivery systems and shall not be diverted to any country, company or individual if prohibited by the applicable export laws of any country. 3DS Offerings may not be used in connection with the design or manufacturing of military items in a country subject to an arms embargo decided by a common position or joint action adopted by the Council of the European Union, or a decision of the Organization for Security and Cooperation in Europe, or an arms embargo imposed by a binding resolution of the Security Council of the United Nations.

Customer recognizes that Customer Data may be transferred to or stored in any country. Customer undertakes to abstain from, and shall ensure all users abstain from, processing, storing or uploading on its data sharing environment any information or data, the export of which is controlled, regulated or subject to any permit or license under any applicable law or regulation. Customer shall be deemed to be the exporter of Customer Data. 3DS may terminate this Agreement and all licenses and access to the Online Services hereunder upon written notice if Customer violates these provisions.

TERMS SPECIFIC TO KENYA

Definitions – In addition to the definition of "Agreement" in Section 1 of the General Terms, the following shall apply:

The terms contained in the 3DS Website Terms referenced herein and hereby incorporated by reference will be provided electronically by 3DS or Distributor.

TERMS SPECIFIC TO LEBANON

■ Taxes – In addition to the provisions of Section 4.2.3 of the General Terms, the following shall apply:

All prices are exclusive of taxes. Customer shall be responsible for payment of any and all taxes, including fees, duties, excises, import VAT, or similar charges of any nature whatsoever, now in force or enacted in the future, that are levied, assessed, charged, withheld, or collected for or in connection with the transfer or usage, to the extent authorized hereunder, of the 3DS Offerings provided hereunder or otherwise arising in connection with this Agreement, but excluding domestic taxes based on 3DS's net income.

If Customer is required to withhold, deduct, or pay for any tax from the amount of fees to be paid under this Agreement, then Customer shall pay such additional amount to 3DS as is necessary to ensure that 3DS receives a sum equal to what would have been received had no such withholding, deduction or payment been required.

Customer shall be responsible for (i) the withholding and payment to the Lebanese Ministry of Finance of non-resident income tax (currently at 7.5%), evidence of such payment signed by the competent Tax Authority must be sent to 3DS as soon as possible, (ii) the withholding and

payment to the Lebanese Ministry of Finance of value-added tax (VAT) (currently at 10%), and (iii) the stamp duty that is or would later become applicable in relation to this Agreement.

TERMS SPECIFIC TO MACEDONIA

Limitation of Liability – The provisions of Section 7 of the General Terms shall be entirely deleted and replaced as follows:

EXCEPT FOR 3DS'S LIABILITY UNDER SECTION 5.2 HEREOF, and liability in case of intention or gross negligence, 3DS'S MAXIMUM LIABILITY FOR DAMAGES SHALL NOT EXCEED THE AMOUNT OF FEES ACTUALLY PAID BY CUSTOMER FOR THE LICENSED PROGRAM OR THE ONLINE SERVICES WHICH CAUSED THE DAMAGES IN THE PRECEDING TWELVE (12) MONTH-PERIOD PRIOR TO THE OCCURRENCE OF THE CAUSE OF ACTION GIVING RISE TO THE CLAIM.

Save for liability in case of intention or gross negligence, 3DS SHALL HAVE NO LIABILITY FOR INDIRECT, INCIDENTAL, CONSEQUENTIAL OR PUNITIVE DAMAGES, INCLUDING WITHOUT LIMITATION CLAIMS FOR LOST PROFITS, BUSINESS INTERRUPTION AND LOSS OF DATA, THAT IN ANY WAY RELATE TO THIS AGREEMENT, ANY 3DS OFFERING, DOCUMENTATION OR SERVICES, WHETHER OR NOT 3DS HAS BEEN ADVISED OF THE POSSIBILITY OF SUCH DAMAGES AND NOTWITHSTANDING THE FAILURE OF THE ESSENTIAL PURPOSE OF ANY REMEDY.

THE LIMITATIONS STATED IN THIS SECTION SHALL APPLY REGARDLESS OF THE FORM OF ACTION, WHETHER THE ASSERTED LIABILITY OR DAMAGES ARE BASED ON CONTRACT (INCLUDING, BUT NOT LIMITED TO, BREACH OF WARRANTY), TORT (INCLUDING, BUT NOT LIMITED TO, NEGLIGENCE), STATUTE, OR ANY OTHER LEGAL OR EQUITABLE THEORY.

Customer waives any and all claims related to this Agreement or any 3DS Offerings or Documentation or services provided hereunder, for any direct, indirect, incidental or consequential damages, on any basis other than in case of intention or gross negligence, against any 3DS licensors or any 3DS Group Company other than 3DS.

Any legal action against 3DS must be filed within the appropriate judicial jurisdiction within the deadlines prescribed for such action under the applicable laws.

TERMS SPECIFIC TO MAURITANIA

Section 4.2.2 of the General Terms and Section "Late Payment" of the terms applicable to Africa are not applicable.

TERMS SPECIFIC TO ROMANIA

Miscellaneous – In addition to the provisions of Section 14 of the General Terms, the following shall apply:

The Customer hereby represents that it has read, understood, expressly and individually agreed and accepted the provisions of Sections 3, 5.2, 6, 7, 9.2, 9.3, 9.4, 10.4, 12, 15.4 and 15.13 of the General Terms and the provisions of Section "Governing Law and Jurisdiction" of the Country Specific Terms.

TERMS SPECIFIC TO SAUDI ARABIA

Late Payments – In addition to the provisions of Section 4.2.2 of the General Terms, the following shall apply:

Customer shall pay interest for late payment from the due date up to the date of actual payment, after as well as before judgment, at the judgment rate or a rate of eight percent (8%) per annum over the Bank of Interest Base Rate, plus reasonable attorneys' fees and costs incurred by 3DS in collecting unpaid amounts.

Taxes – In addition to the provisions of Section 4.2.3 of the General Terms, the following shall apply:

All prices are exclusive of taxes. Customer shall be responsible for payment of any and all taxes, including fees, duties, excises, import VAT, or similar charges of any nature whatsoever, now in force or enacted in the future, that are levied, assessed, charged, withheld, or collected for or

in connection with the transfer or usage, to the extent authorized hereunder, of the 3DS Offerings provided hereunder or otherwise arising in connection with this Agreement, but excluding domestic taxes based on 3DS's net income.

If Customer is required to withhold, deduct, or pay for any tax from the amount of fees to be paid under this Agreement, then Customer shall pay such additional amount to 3DS as is necessary to ensure that 3DS receives a sum equal to what would have been received had no such withholding, deduction or payment been required. Customer shall comply with its local obligations, including but not limited to registration with the Department of Zakat and filing of annual withholding tax returns. Customer shall indemnify 3DS against any losses or costs incurred by 3DS due to any failure of Customer to make such deduction or withholding.

Limitation of Liability – The provisions of Section 7 of the General Terms shall be entirely deleted and replaced as follows:

EXCEPT FOR 3DS'S LIABILITY UNDER SECTION 5.2 HEREOF, 3DS'S MAXIMUM LIABILITY FOR DAMAGES SHALL NOT EXCEED THE AMOUNT OF FEES ACTUALLY PAID BY CUSTOMER FOR THE LICENSED PROGRAM OR THE ONLINE SERVICES WHICH CAUSED THE DAMAGES IN THE PRECEDING TWELVE (12) MONTH-PERIOD PRIOR TO THE OCCURRENCE OF THE CAUSE OF ACTION GIVING RISE TO THE CLAIM, PROVIDED THAT IF LESS THAN TWELVE (12) MONTHS HAS ELAPSED FROM THE EFFECTIVE DATE OF THE LICENSED PROGRAM OR THE ONLINE SERVICES, AS APPLICABLE, TO THE DATE OF SUCH OCCURRENCE, 3DS'S MAXIMUM LIABILITY SHALL BE THE FEES PAYABLE IN THE FIRST TWELVE (12) MONTHS FOLLOWING THE EFFECTIVE DATE FOR THE LICENSED PROGRAM OR THE ONLINE SERVICES WHICH CAUSED THE DAMAGE, AS APPLICABLE,

3DS SHALL HAVE NO LIABILITY FOR INDIRECT, INCIDENTAL, CONSEQUENTIAL, SPECIAL OR PUNITIVE DAMAGES, WHETHER OR NOT 3DS HAS BEEN ADVISED OF THE POSSIBILITY OF SUCH DAMAGES.

3DS SHALL HAVE NO LIABILITY FOR ANY AND ALL CLAIMS FOR LOSS OF PROFITS, LOSS OF ANTICIPATED SAVINGS, LOSS OF BUSINESS OPPORTUNITY, LOSS OF GOODWILL, LOSS OF DATA (SAVE ANY DATA STORED BY 3DS AS PART OF THE ONLINE SERVICES AND LOST DUE TO 3DS'S NEGLIGENCE, AND IN ANY CASE ONLY FOR SUCH DATA AS COULD NOT HAVE BEEN REASONABLY BACKED UP BY CUSTOMER), IN EACH CASE, WHETHER THE SAME ARE SUFFERED DIRECTLY OR INDIRECTLY OR ARE IMMEDIATE OR CONSEQUENTIAL, ARISING OUT OF OR IN CONNECTION WITH THIS AGREEMENT, THE 3DS OFFERINGS, THE DOCUMENTATION OR SERVICES.

THE LIMITATIONS STATED IN THIS SECTION SHALL APPLY REGARDLESS OF THE FORM OF ACTION, WHETHER THE ASSERTED LIABILITY OR DAMAGES ARE BASED ON CONTRACT (INCLUDING, BUT NOT LIMITED TO, BREACH OF WARRANTY), TORT (INCLUDING, BUT NOT LIMITED TO, NEGLIGENCE), STATUTE, OR ANY OTHER LEGAL OR EQUITABLE THEORY.

IN ALL CASES PROVIDED, HOWEVER, THAT NOTHING IN THIS AGREEMENT SHALL LIMIT OR EXCLUDE 3DS'S LIABILITY FOR DEATH OR PERSONAL INJURY CAUSED BY THE NEGLIGENCE OF 3DS, ITS OFFICERS, EMPLOYEES, CONTRACTORS OR AGENTS; FOR FRAUD OR FRAUDULENT MISREPRESENTATION; FOR BREACH OF THE OBLIGATIONS IMPLIED BY SECTION 12 SALE OF GOODS ACT 1979 OR SECTION 2 SUPPLY OF GOODS AND SERVICES ACT 1982 EACH IN ENGLAND AND WALES; OR FOR ANY OTHER LIABILITY WHICH MAY NOT BE EXCLUDED BY LAW.

BY ENTERING INTO THIS AGREEMENT, CUSTOMER AGREES AND ACKNOWLDGES THAT THE LIMITATIONS AND EXCLUSIONS OF LIABILITY HERE CONTAINED ARE REASONABLE IN VIEW OF THE FEES PAYABLE BY CUSTOMER.

IN ENTERING INTO THIS AGREEMENT, CUSTOMER HAS NOT RELIED ON ANY REPRESENTATIONS NOT SET OUT IN THIS AGREEMENT AND 3DS'S LIABILITY IS EXCLUDED FOR ALL REPRESENTATIONS NOT HERE SET OUT, SUBJECT AS ABOVE PROVIDED.

Customer waives any and all claims related to this Agreement or any 3DS Offerings or Documentation or services provided hereunder, for any direct, incidental or consequential damages, on any basis, against any 3DS licensors or any 3DS Group Company other than 3DS.

Any legal action against 3DS must be filed with the appropriate judicial jurisdiction within two (2) years after the applicable cause of action has arisen.

Governing Law and Jurisdiction – In addition to the provisions of Section 14.11 of the General Terms, the following shall apply:

The Agreement shall be governed and construed in accordance with the laws of England and Wales. All disputes arising out of or in connection with this Agreement shall be finally settled under the Rules of Arbitration of the International Chamber of Commerce by one or more arbitrators appointed in accordance with the said Rules of Arbitration. The seat of the arbitration shall be in Paris, France. The arbitral proceedings shall be

conducted in English. Any party to this Agreement shall have the right to have recourse to and shall be bound by the Pre-Arbitral Referee Procedure of the International Chamber of Commerce in accordance with its Rules for a Pre-Arbitral Referee Procedure. Moreover, either party is entitled to seek at any time interim relief (including injunctive relief) before any State court of competent jurisdiction.

TERMS SPECIFIC TO SERBIA

• Intellectual Property Claims – The provisions of Section 5.2 of the General Terms shall be entirely deleted and replaced as follows:

3DS will defend Customer against any claims made by a third party that a 3DS Offering delivered under this Agreement, inter alia, against claims on infringement of a copyright in any country or a patent of the United States, Japan, or a member state of the European Patent Organization, and will pay all costs, damages and expenses (including reasonable legal fees) finally awarded against Customer by a court of competent jurisdiction or agreed to in a written settlement agreement signed by 3DS arising out of such claim, provided (i) Customer provides 3DS with prompt written notice of the claim, and (ii) Customer gives 3DS sole control of the defense of the claim and any related settlement discussions and provides reasonable cooperation in the defense and settlement of the claim.

If such a claim is made, or in 3DS's reasonable opinion a 3DS Offering is likely to become the subject of such a claim (or likely to be made), 3DS may at 3DS's expense, either secure the right for Customer to continue using the 3DS applicable Offering, modify it so that it is not infringing, or replace it with another program which is functionally equivalent. If none of the foregoing options is available on terms which are reasonable in 3DS's judgment, 3DS may terminate the licenses to the Licensed Program and/or terminate the Online Services. For perpetual licenses, 3DS shall either refund or provide a credit to Customer, at Customer's option, in an amount equal to the corresponding one-time fee paid for the licenses, depreciated on a straight-line over three (3) years upon return or destruction of all copies of the affected Licensed Program as certified by an officer of Customer. For Online Services or licenses that are not perpetual, 3DS shall refund all prepaid but unused fees paid hereunder for the affected Online Services or licenses.

3DS shall have no obligation to defend or indemnify Customer against any claim related to (i) any modification of a 3DS Offering by anybody other than 3DS, (ii) the use of one or more 3DS Offerings in combination with other hardware, data or programs not specified by 3DS, or (iii) the use of corrective patches or Releases other than the most recent one.

This Section 5.2 states 3DS's entire liability and Customer's exclusive remedy for any claim of infringement of intellectual property rights.

Disclaimers – The provisions of Section 6.3 of the General Terms shall be entirely deleted and replaced as follows:

THE FOREGOING WARRANTIES ARE IN LIEU OF AND EXCLUDE ALL OTHER WARRANTIES, REPRESENTATIONS OR CONDITIONS FOR 3DS OFFERINGS, WHETHER EXPRESS OR IMPLIED, INCLUDING, WITHOUT LIMITATION, ANY IMPLIED WARRANTIES OF MERCHANTABILITY, FITNESS FOR A PARTICULAR PURPOSE, TITLE OR NON INFRINGEMENT.

3DS disclaims all liability for any use or application of any 3DS Offering or the results or decisions made or obtained by users of the 3DS Offering. 3DS does not warrant that (i) the functions of any 3DS Offering will meet Customer's requirements or will enable it to attain the objectives Customer has set for itself, or (ii) the 3DS Offering will operate in the combination or environment selected for use by Customer. In all instances, Customer shall be solely responsible for ensuring that the results produced by 3DS Offering comply with quality and safety requirements of Customer's products or services. No employee or agent of 3DS is authorized to give a greater or different warranty. Customer shall have exclusive responsibility for (a) selection of the 3DS Offering to achieve Customer's intended results, (b) installation of the Licensed Program, (c) taking adequate measures to properly test, operate and use each 3DS Offering, and (d) results obtained therefrom.

The disclaimers above apply to the maximum extent permitted by applicable law.

Third Party Hosting – The provisions of Section 14.3 of the General Terms shall be entirely deleted and replaced as follows:

Customer is authorized to install and use the Licensed Programs remotely on computers operated by a well-established, reputable third party service provider, and to appoint such service provider to operate the hardware and manage the Licensed Programs solely for and on behalf of Customer; provided however, that (i) only duly authorized users shall have the right to use the Licensed Programs; and (ii) Customer shall enter into a written agreement with such service provider under which the service provider agrees that its access to the Licensed Programs is solely for the purpose of providing the services mentioned above to Customer and is otherwise subject to all of the restrictions and limitations contained in this Agreement. Customer acknowledges and agrees that the service provider shall be deemed an agent of Customer. If Customer becomes aware of any actual or suspected unauthorized access, use or disclosure of the Licensed Programs, Customer shall immediately terminate the service provider's access to the Licensed Programs. Customer shall defend and indemnify 3DS against any claim, expense, judgment, damage

or loss (including reasonable attorneys' fees), which arises out of or in any way relates to any such service provider's access to or use of the Licensed Programs.

TERMS SPECIFIC TO SLOVAKIA

■ Late Payments – In addition to the provisions of Section 4.2.3 of the General Terms, the following shall apply:

Customer shall pay interest for late payment at a rate of twelve (12) percent per year on all sums unpaid at the due date, plus a lump sum compensation for recovery costs in amount of forty (40) Euros. The right of 3DS to claim damages from the Customer caused by the late payment, to the extent it is not covered by the late payment interest and lump sum compensation, and the right of 3DS to claim legal costs (including court fees) necessary for enforcement of the 3DS's claim, remain intact.

TERMS SPECIFIC TO TANZANIA

• Limitation of Liability – The last provision of Section 7 of the General Terms shall be amended as follows:

To the extent permitted by any applicable law, any legal action against 3DS must be filed with the appropriate judicial jurisdiction within two (2) years after the applicable cause of action has arisen.

TERMS SPECIFIC TO UNITED ARAB EMIRATES

■ Late Payments – In addition to the provisions of Section 4.2.3 of the General Terms, the following shall apply:

Customer shall pay interest for late payment from the due date up to the date of actual payment, after as well as before judgment, at the judgment rate or a rate of eight percent (8%) per annum over the Bank of Interest Base Rate, plus reasonable attorneys' fees and costs incurred by 3DS in collecting unpaid amounts.

Limitation of Liability – The provisions of Section 7 of the General Terms shall be entirely deleted and replaced as follows:

EXCEPT FOR 3DS'S LIABILITY UNDER SECTION 5.2 HEREOF, 3DS'S MAXIMUM LIABILITY FOR DAMAGES SHALL NOT EXCEED THE AMOUNT OF FEES ACTUALLY PAID BY CUSTOMER FOR THE LICENSED PROGRAM OR THE ONLINE SERVICES WHICH CAUSED THE DAMAGES IN THE PRECEDING TWELVE (12) MONTH-PERIOD PRIOR TO THE OCCURRENCE OF THE CAUSE OF ACTION GIVING RISE TO THE CLAIM, PROVIDED THAT IF LESS THAN TWELVE (12) MONTHS HAS ELAPSED FROM THE EFFECTIVE DATE OF THE LICENSED PROGRAM OR THE ONLINE SERVICES, AS APPLICABLE, TO THE DATE OF SUCH OCCURRENCE, 3DS'S MAXIMUM LIABILITY SHALL BE THE FEES PAYABLE IN THE FIRST TWELVE (12) MONTHS FOLLOWING THE EFFECTIVE DATE FOR THE LICENSED PROGRAM OR THE ONLINE SERVICES WHICH CAUSED THE DAMAGE, AS APPLICABLE,

3DS SHALL HAVE NO LIABILITY FOR INDIRECT, INCIDENTAL, CONSEQUENTIAL, SPECIAL OR PUNITIVE DAMAGES, WHETHER OR NOT 3DS HAS BEEN ADVISED OF THE POSSIBILITY OF SUCH DAMAGES.

3DS SHALL HAVE NO LIABILITY FOR ANY AND ALL CLAIMS FOR LOSS OF PROFITS, LOSS OF ANTICIPATED SAVINGS, LOSS OF BUSINESS OPPORTUNITY, LOSS OF GOODWILL, LOSS OF DATA (SAVE ANY DATA STORED BY 3DS AS PART OF THE ONLINE SERVICES AND LOST DUE TO 3DS'S NEGLIGENCE, AND IN ANY CASE ONLY FOR SUCH DATA AS COULD NOT HAVE BEEN REASONABLY BACKED UP BY CUSTOMER), IN EACH CASE, WHETHER THE SAME ARE SUFFERED DIRECTLY OR INDIRECTLY OR ARE IMMEDIATE OR CONSEQUENTIAL, ARISING OUT OF OR IN CONNECTION WITH THIS AGREEMENT, THE 3DS OFFERINGS, THE DOCUMENTATION OR SERVICES.

THE LIMITATIONS STATED IN THIS SECTION SHALL APPLY REGARDLESS OF THE FORM OF ACTION, WHETHER THE ASSERTED LIABILITY OR DAMAGES ARE BASED ON CONTRACT (INCLUDING, BUT NOT LIMITED TO, BREACH OF WARRANTY), TORT (INCLUDING, BUT NOT LIMITED TO, NEGLIGENCE), STATUTE, OR ANY OTHER LEGAL OR EQUITABLE THEORY.

IN ALL CASES PROVIDED, HOWEVER, THAT NOTHING IN THIS AGREEMENT SHALL LIMIT OR EXCLUDE 3DS'S LIABILITY FOR DEATH OR PERSONAL INJURY CAUSED BY THE NEGLIGENCE OF 3DS, ITS OFFICERS, EMPLOYEES, CONTRACTORS OR AGENTS; FOR FRAUD OR FRAUDULENT MISREPRESENTATION; FOR BREACH OF THE OBLIGATIONS IMPLIED BY SECTION 12 SALE OF GOODS ACT

1979 OR SECTION 2 SUPPLY OF GOODS AND SERVICES ACT 1982 EACH IN ENGLAND AND WALES; OR FOR ANY OTHER LIABILITY WHICH MAY NOT BE EXCLUDED BY LAW.

BY ENTERING INTO THIS AGREEMENT, CUSTOMER AGREES AND ACKNOWLDGES THAT THE LIMITATIONS AND EXCLUSIONS OF LIABILITY HERE CONTAINED ARE REASONABLE IN VIEW OF THE FEES PAYABLE BY CUSTOMER.

IN ENTERING INTO THIS AGREEMENT, CUSTOMER HAS NOT RELIED ON ANY REPRESENTATIONS NOT SET OUT IN THIS AGREEMENT AND 3DS'S LIABILITY IS EXCLUDED FOR ALL REPRESENTATIONS NOT HERE SET OUT, SUBJECT AS ABOVE PROVIDED.

Customer waives any and all claims related to this Agreement or any 3DS Offerings or Documentation or services provided hereunder, for any direct, indirect, incidental or consequential damages, on any basis, against any 3DS licensors or any 3DS Group Company other than 3DS.

Any legal action against 3DS must be filed with the appropriate judicial jurisdiction within two (2) years after the applicable cause of action has arisen.

Governing Law and Jurisdiction – In addition to the provisions of Section 14.11 of the General Terms, the following shall apply:

The Agreement shall be governed and construed in accordance with the laws of England and Wales. All disputes arising out of or in connection with this Agreement shall be finally settled under the Rules of Arbitration of the International Chamber of Commerce by one or more arbitrators appointed in accordance with the said Rules of Arbitration. The seat of the arbitration shall be in Paris, France. The arbitral proceedings shall be conducted in English. Any party to this Agreement shall have the right to have recourse to and shall be bound by the Pre-Arbitral Referee Procedure of the International Chamber of Commerce in accordance with its Rules for a Pre-Arbitral Referee Procedure. Moreover, either party is entitled to seek at any time interim relief (including injunctive relief) before any State court of competent jurisdiction.

TERMS SPECIFIC TO UNITED KINGDOM

■ Late Payments – In addition to the provisions of Section 4.2.2 of the General Terms, the following shall apply:

Customer shall pay interest for late payment from the due date up to the date of actual payment, after as well as before judgment, at the judgment rate or a rate of eight percent (8%) per annum over the Bank of Interest Base Rate, plus reasonable attorneys' fees and costs incurred by 3DS in collecting unpaid amounts.

Limitation of Liability – In addition to the provisions of Section 7 of the General Terms, the following shall apply:

Nothing in the Agreement excludes or limits any liability which cannot be excluded or limited as a matter of law under the laws of England and Wales, including but not limited to liability for: (a) death or personal injury caused by negligence; (b) fraud or fraudulent misrepresentation; and (c) breach of the terms implied by section 12 of the Sale of Goods Act 1979 or section 2 of the Supply of Goods and Services Act 1982 (title and quiet possession).

TERMS SPECIFIC TO ZAMBIA

• Governing Law and Jurisdiction – In addition to the provisions of Section 14.11 of the General Terms, the following shall apply:

Any judgment or order made by the Commercial Court of Paris ("Tribunal de Commerce de Paris") may be registered in the High Court of the Republic of Zambia and the Customer hereby consents to such registration.

CLOSA (Indirect) - EMEA - V13.2.1